Author's response to reviews

Title: The perception of primary health care professionals about education in diabetes melitus, Brazil

Authors:

   Heloisa C Torres (heloisa@enf.ufmg.br)
   Marta A Amaral (marta@enf.ufmg.br)
   Brani R Rosemberg (Brani@emsp.fiocruz.br)
   Regina C Bodstein (bodstein@ensp.fiocruz.br)

Version: 2 Date: 1 February 2010

Author's response to reviews:

none